Author's response to reviews

Title: Porn video shows, local brew, and transactional sex: HIV risk among youth in Kisumu, Kenya

Authors:

Carolyne Njue (ngari.w@gmail.com)
Helene ACM Voeten (H.Voeten@ggd.rotterdam.nl)
Pieter Remes (premes@gmail.com)

Version: 7 Date: 6 August 2011

Author's response to reviews: see over
We have checked the manuscript further and made a few revisions to the references section.

Reference 8: Had overlooked the changes, so revised reference to include all the authors.


Typography: We have checked the manuscript for typographical errors and made small changes (commas, full-stop).

Yours sincerely, also on behalf of my co-authors,

Carolyne Njue